Miscommunication and Clarification Requests (Was: Miscommunication and Repair Strategies in Dialogue)

Malte Gabsdil

Universität des Saarlandes

EGK Colloquium January 28,2002 – p.1/24

Overview

All in one (What has changed since Bostalsee?) Repairs and Clarification Requests Previous Work Baufix Corpus Ingredients (Miscommunication, Grounding, Focus, Questions) Summary What to do next?

EGK Colloquium January 28,2002 – p.2/24

All in one (What has changed since Bostalsee?)

EGK Colloquium January 28,2002 – p.3/24

All in one (What has changed since Bostalsee?)

The title

EGK Colloquium January 28,2002 – p.3/24

All in one (What has changed since Bostalsee?)

The title Refinement/Reformulation of terms Miscommunication part ok; preliminary work on an extension of a proposed classification scheme The notion of “Repair Strategies” was (almost) entirely void

EGK Colloquium January 28,2002 – p.3/24

All in one (What has changed since Bostalsee?)

The title Refinement/Reformulation of terms Miscommunication part ok; preliminary work on an extension of a proposed classification scheme The notion of “Repair Strategies” was (almost) entirely void Paraphrase for “How do people deal with misunderstandings” with a focus on corrections

EGK Colloquium January 28,2002 – p.3/24

All in one (What has changed since Bostalsee?)

The title Refinement/Reformulation of terms

Miscommunication part ok; preliminary work on an extension of a proposed classification scheme The notion of “Repair Strategies” was (almost) entirely void Paraphrase for “How do people deal with misunderstandings” with a focus on corrections Corrections are rare and only one (dispreferred) means for dealing with a special type of misunderstandings

EGK Colloquium January 28,2002 – p.3/24

All in one (What has changed since Bostalsee?)

The title Refinement/Reformulation of terms

Miscommunication part ok; preliminary work on an extension of a proposed classification scheme The notion of “Repair Strategies” was (almost) entirely void Paraphrase for “How do people deal with misunderstandings” with a focus on corrections Corrections are rare and only one (dispreferred) means for dealing with a special type of misunderstandings Much more common are clarification requests

EGK Colloquium January 28,2002 – p.3/24

Repairs (I)

What do people usually mean by repairing miscommunication?

EGK Colloquium January 28,2002 – p.4/24

Repairs (I)

What do people usually mean by repairing miscommunication? Hirst et al. ’94: “Utterance that attempts to correct the problem.”

EGK Colloquium January 28,2002 – p.4/24

Repairs (I)

What do people usually mean by repairing miscommunication? Hirst et al. ’94: “Utterance that attempts to correct the problem.” Traum ’94: “. . . attempts to fix previous utterances that are perceived to be (possibly) insufficient for conveying what was intended.” Repairs

Clarifications Corrections

EGK Colloquium January 28,2002 – p.4/24

Repairs (II)

Repairs (in Traum’s sense) can be classified according to

EGK Colloquium January 28,2002 – p.5/24

Repairs (II)

Repairs (in Traum’s sense) can be classified according to who made them (self or other)

EGK Colloquium January 28,2002 – p.5/24

Repairs (II)

Repairs (in Traum’s sense) can be classified according to who made them (self or other) initiated them (self or other)

EGK Colloquium January 28,2002 – p.5/24

Repairs (II)

Repairs (in Traum’s sense) can be classified according to who made them (self or other) initiated them (self or other) how far they are away from the utterance they repair

EGK Colloquium January 28,2002 – p.5/24

Repairs (II)

Repairs (in Traum’s sense) can be classified according to who made them (self or other) initiated them (self or other) how far they are away from the utterance they repair Preference scheme for repairs (Schegloff ’77): other-corrections dispreferred

EGK Colloquium January 28,2002 – p.5/24

Repairs (II)

Repairs (in Traum’s sense) can be classified according to who made them (self or other) initiated them (self or other) how far they are away from the utterance they repair Preference scheme for repairs (Schegloff ’77): other-corrections dispreferred Initiation of repair by other conversant with Next Turn Repair Initiator (NTRI) → clarifictaion request

EGK Colloquium January 28,2002 – p.5/24

Clarification Requests

→ important device for handling miscommunication

EGK Colloquium January 28,2002 – p.6/24

Clarification Requests

Communication is a collaborative process (e.g. Clark ’96)

→ important device for handling miscommunication

EGK Colloquium January 28,2002 – p.6/24

Clarification Requests

Communication is a collaborative process (e.g. Clark ’96) mutual responsibility for understanding of each utterance minimize joint effort (efficiency)

→ important device for handling miscommunication

EGK Colloquium January 28,2002 – p.6/24

Clarification Requests

Communication is a collaborative process (e.g. Clark ’96) mutual responsibility for understanding of each utterance minimize joint effort (efficiency) Clarification Requests

→ important device for handling miscommunication

EGK Colloquium January 28,2002 – p.6/24

Clarification Requests

Communication is a collaborative process (e.g. Clark ’96) mutual responsibility for understanding of each utterance minimize joint effort (efficiency) Clarification Requests

indicate “problematic” (non or partially understood) utterances help minimizing joint effort

→ important device for handling miscommunication

EGK Colloquium January 28,2002 – p.6/24

Clarification Requests

Communication is a collaborative process (e.g. Clark ’96) mutual responsibility for understanding of each utterance minimize joint effort (efficiency) Clarification Requests

indicate “problematic” (non or partially understood) utterances help minimizing joint effort

→ important device for handling miscommunication

EGK Colloquium January 28,2002 – p.6/24

Clarification Requests

Communication is a collaborative process (e.g. Clark ’96) mutual responsibility for understanding of each utterance minimize joint effort (efficiency) Clarification Requests

indicate “problematic” (non or partially understood) utterances help minimizing joint effort

→ important device for handling miscommunication

EGK Colloquium January 28,2002 – p.6/24

Previous Work

EGK Colloquium January 28,2002 – p.7/24

Previous Work

Important References (general)

EGK Colloquium January 28,2002 – p.7/24

Previous Work

Important References (general) Perlis et al. ’98: Meta-linguistic abilities essential for successful communication (Clark: talk about talk is also talk)

EGK Colloquium January 28,2002 – p.7/24

Previous Work

Important References (general) Perlis et al. ’98: Meta-linguistic abilities essential for successful communication (Clark: talk about talk is also talk) Clark & Wilkes-Gibbs ’86: Communication as a collaborative process

EGK Colloquium January 28,2002 – p.7/24

Previous Work

Important References (general) Perlis et al. ’98: Meta-linguistic abilities essential for successful communication (Clark: talk about talk is also talk) Clark & Wilkes-Gibbs ’86: Communication as a collaborative process Traum ’94, Clark ’96: Grounding models, mutual understanding

EGK Colloquium January 28,2002 – p.7/24

Previous Work

Important References (general) Perlis et al. ’98: Meta-linguistic abilities essential for successful communication (Clark: talk about talk is also talk) Clark & Wilkes-Gibbs ’86: Communication as a collaborative process Traum ’94, Clark ’96: Grounding models, mutual understanding Important References (specific)

EGK Colloquium January 28,2002 – p.7/24

Previous Work

Important References (general) Perlis et al. ’98: Meta-linguistic abilities essential for successful communication (Clark: talk about talk is also talk) Clark & Wilkes-Gibbs ’86: Communication as a collaborative process Traum ’94, Clark ’96: Grounding models, mutual understanding Important References (specific) Ginzburg ’98; G & Cooper ’01; G & Cooper (forthcoming); G & Sag ’00; G, Gregory & Lappin ’01; Purver, G & Healey ’01

EGK Colloquium January 28,2002 – p.7/24

Purver, Ginzburg & Healey (PGH) ’01

EGK Colloquium January 28,2002 – p.8/24

Purver, Ginzburg & Healey (PGH) ’01

Empirical study of clarification phenomena (esp. syntactic form)

EGK Colloquium January 28,2002 – p.8/24

Purver, Ginzburg & Healey (PGH) ’01

Empirical study of clarification phenomena (esp. syntactic form) Corpus: parts of BNC (200-speaker-turn sections from 59 transcripts; approx. 150.000 words)

EGK Colloquium January 28,2002 – p.8/24

Purver, Ginzburg & Healey (PGH) ’01

Empirical study of clarification phenomena (esp. syntactic form) Corpus: parts of BNC (200-speaker-turn sections from 59 transcripts; approx. 150.000 words) 4% of sentences clarification requests (of these 29% clarification ellipsis) Mark clarification requests for form, meaning, and (source)

EGK Colloquium January 28,2002 – p.8/24

Purver, Ginzburg & Healey (PGH) ’01

Empirical study of clarification phenomena (esp. syntactic form) Corpus: parts of BNC (200-speaker-turn sections from 59 transcripts; approx. 150.000 words) 4% of sentences clarification requests (of these 29% clarification ellipsis) Mark clarification requests for form, meaning, and (source) Form/Reading Distribution, (Source Separation Distance)

EGK Colloquium January 28,2002 – p.8/24

PGH ’01: Clarification Forms

EGK Colloquium January 28,2002 – p.9/24

PGH ’01: Clarification Forms

Non-reprise clarifications (explicitly ask for clarification)

EGK Colloquium January 28,2002 – p.9/24

PGH ’01: Clarification Forms

Non-reprise clarifications (explicitly ask for clarification) Reprise sentences (echoing/repeating a previous utterance in full)

EGK Colloquium January 28,2002 – p.9/24

PGH ’01: Clarification Forms

Non-reprise clarifications (explicitly ask for clarification) Reprise sentences (echoing/repeating a previous utterance in full) Reprise sluices (elliptical wh-constructions which receive a “sentential” interpretation)

EGK Colloquium January 28,2002 – p.9/24

PGH ’01: Clarification Forms

Non-reprise clarifications (explicitly ask for clarification) Reprise sentences (echoing/repeating a previous utterance in full) Reprise sluices (elliptical wh-constructions which receive a “sentential” interpretation) Reprise fragments (elliptical bare fragments)

EGK Colloquium January 28,2002 – p.9/24

PGH ’01: Clarification Forms

Non-reprise clarifications (explicitly ask for clarification) Reprise sentences (echoing/repeating a previous utterance in full) Reprise sluices (elliptical wh-constructions which receive a “sentential” interpretation) Reprise fragments (elliptical bare fragments) Gaps (repeat preceding word)

EGK Colloquium January 28,2002 – p.9/24

PGH ’01: Clarification Forms

Non-reprise clarifications (explicitly ask for clarification) Reprise sentences (echoing/repeating a previous utterance in full) Reprise sluices (elliptical wh-constructions which receive a “sentential” interpretation) Reprise fragments (elliptical bare fragments) Gaps (repeat preceding word) Gap fillers (fill previous incomplete utterance)

EGK Colloquium January 28,2002 – p.9/24

PGH ’01: Clarification Forms

Non-reprise clarifications (explicitly ask for clarification) Reprise sentences (echoing/repeating a previous utterance in full) Reprise sluices (elliptical wh-constructions which receive a “sentential” interpretation) Reprise fragments (elliptical bare fragments) Gaps (repeat preceding word) Gap fillers (fill previous incomplete utterance) Conventional (indicate complete breakdown)

EGK Colloquium January 28,2002 – p.9/24

PGH ’01: Clarification Readings

EGK Colloquium January 28,2002 – p.10/24

PGH ’01: Clarification Readings

Clausal (confirmation)

EGK Colloquium January 28,2002 – p.10/24

PGH ’01: Clarification Readings

Clausal (confirmation) “Are you asking/asserting that X?” “For which X are you asking/asserting that Y”

EGK Colloquium January 28,2002 – p.10/24

PGH ’01: Clarification Readings

Clausal (confirmation) “Are you asking/asserting that X?” “For which X are you asking/asserting that Y” Constituent (reference)

EGK Colloquium January 28,2002 – p.10/24

PGH ’01: Clarification Readings

Clausal (confirmation) “Are you asking/asserting that X?” “For which X are you asking/asserting that Y” Constituent (reference) “What/who is X?”

“What/who do you mean by X?”

EGK Colloquium January 28,2002 – p.10/24

PGH ’01: Clarification Readings

Clausal (confirmation) “Are you asking/asserting that X?” “For which X are you asking/asserting that Y” Constituent (reference) “What/who is X?”

“What/who do you mean by X?” Lexical (surface)

EGK Colloquium January 28,2002 – p.10/24

PGH ’01: Clarification Readings

Clausal (confirmation) “Are you asking/asserting that X?” “For which X are you asking/asserting that Y” Constituent (reference) “What/who is X?”

“What/who do you mean by X?” Lexical (surface)

“Did you utter X?” “What did you utter?”

EGK Colloquium January 28,2002 – p.10/24

PGH ’01: Clarification Readings

Clausal (confirmation) “Are you asking/asserting that X?” “For which X are you asking/asserting that Y” Constituent (reference) “What/who is X?”

“What/who do you mean by X?” Lexical (surface)

“Did you utter X?” “What did you utter?”

(Correction)

EGK Colloquium January 28,2002 – p.10/24

Ginzburg & Cooper ’01

EGK Colloquium January 28,2002 – p.11/24

Ginzburg & Cooper ’01

Interpretation of clarification ellipsis in a HPSG framework

EGK Colloquium January 28,2002 – p.11/24

Ginzburg & Cooper ’01

Interpretation of clarification ellipsis in a HPSG framework Phonological/syntactic/semantic parallelism → sign-based approach

EGK Colloquium January 28,2002 – p.11/24

Ginzburg & Cooper ’01

Interpretation of clarification ellipsis in a HPSG framework Phonological/syntactic/semantic parallelism → sign-based approach Clarification Potential: Space of possible requests for clarification of a given utterance

EGK Colloquium January 28,2002 – p.11/24

Ginzburg & Cooper ’01

Interpretation of clarification ellipsis in a HPSG framework Phonological/syntactic/semantic parallelism → sign-based approach Clarification Potential: Space of possible requests for clarification of a given utterance Derive possible requests by coercion operations (rules) on utterance representations (HPSG)

EGK Colloquium January 28,2002 – p.11/24

Ginzburg & Cooper ’01

Interpretation of clarification ellipsis in a HPSG framework Phonological/syntactic/semantic parallelism → sign-based approach Clarification Potential: Space of possible requests for clarification of a given utterance Derive possible requests by coercion operations (rules) on utterance representations (HPSG) input utterance → coercion operation → clarification skeleton (must unify with valid clarification)

EGK Colloquium January 28,2002 – p.11/24

Ginzburg & Cooper ’01

Interpretation of clarification ellipsis in a HPSG framework Phonological/syntactic/semantic parallelism → sign-based approach Clarification Potential: Space of possible requests for clarification of a given utterance Derive possible requests by coercion operations (rules) on utterance representations (HPSG) input utterance → coercion operation → clarification skeleton (must unify with valid clarification) Coercion operations part of dialogue competence

EGK Colloquium January 28,2002 – p.11/24

Baufix Corpus (I)

EGK Colloquium January 28,2002 – p.12/24

Baufix Corpus (I)

German task-oriented dialogues recorded at the U. of Bielefeld

EGK Colloquium January 28,2002 – p.12/24

Baufix Corpus (I)

German task-oriented dialogues recorded at the U. of Bielefeld Exp. setup similar to MapTask

EGK Colloquium January 28,2002 – p.12/24

Baufix Corpus (I)

German task-oriented dialogues recorded at the U. of Bielefeld Exp. setup similar to MapTask Constructor has to build a toy-airplane on the basis of instructions

EGK Colloquium January 28,2002 – p.12/24

Baufix Corpus (I)

German task-oriented dialogues recorded at the U. of Bielefeld Exp. setup similar to MapTask Constructor has to build a toy-airplane on the basis of instructions Full of clarification requests:

Dialogue# 1 2 3 4 5 Duration (30.31) (16:10) 9:31 12:15 18:57 Clar. Req. 26 10 9 25 24

EGK Colloquium January 28,2002 – p.12/24

Baufix Corpus (I)

German task-oriented dialogues recorded at the U. of Bielefeld Exp. setup similar to MapTask Constructor has to build a toy-airplane on the basis of instructions Full of clarification requests:

Dialogue# 1 2 3 4 5 Duration (30.31) (16:10) 9:31 12:15 18:57 Clar. Req. 26 10 9 25 24

So far only transcripts but sound-files are on the way (thanks to Hannes Rieser)

EGK Colloquium January 28,2002 – p.12/24

Baufix Corpus (II)

EGK Colloquium January 28,2002 – p.13/24

Baufix Corpus (II)

Clarification requests for ambiguous input (e.g. referential ambiguities)

EGK Colloquium January 28,2002 – p.13/24

Baufix Corpus (II)

Clarification requests for ambiguous input (e.g. referential ambiguities) “Non-linguistic” clarification requests (e.g. action clarifications; how-questions)

EGK Colloquium January 28,2002 – p.13/24

Baufix Corpus (II)

Clarification requests for ambiguous input (e.g. referential ambiguities) “Non-linguistic” clarification requests (e.g. action clarifications; how-questions)

Refer to abstract object, the plan which represents the action that has to be performed

EGK Colloquium January 28,2002 – p.13/24

Baufix Corpus (II)

Clarification requests for ambiguous input (e.g. referential ambiguities) “Non-linguistic” clarification requests (e.g. action clarifications; how-questions)

Refer to abstract object, the plan which represents the action that has to be performed Kind of reference clarification? (Referent of action/part of plan is unknown)

EGK Colloquium January 28,2002 – p.13/24

Baufix Corpus (II)

Clarification requests for ambiguous input (e.g. referential ambiguities) “Non-linguistic” clarification requests (e.g. action clarifications; how-questions)

Refer to abstract object, the plan which represents the action that has to be performed Kind of reference clarification? (Referent of action/part of plan is unknown) Sequences of clarification requests

EGK Colloquium January 28,2002 – p.13/24

Ingredients

EGK Colloquium January 28,2002 – p.14/24

Ingredients

Clarification requests are a dialogue phenomena that involves more than one utterance by more than one speaker

EGK Colloquium January 28,2002 – p.14/24

Ingredients

Clarification requests are a dialogue phenomena that involves more than one utterance by more than one speaker Look at clarifications from “dialogic” and “classical linguistic” viewpoints

EGK Colloquium January 28,2002 – p.14/24

Ingredients

Clarification requests are a dialogue phenomena that involves more than one utterance by more than one speaker Look at clarifications from “dialogic” and “classical linguistic” viewpoints The following topics have to be investigated:

EGK Colloquium January 28,2002 – p.14/24

Ingredients

Clarification requests are a dialogue phenomena that involves more than one utterance by more than one speaker Look at clarifications from “dialogic” and “classical linguistic” viewpoints The following topics have to be investigated:

Miscommunication

EGK Colloquium January 28,2002 – p.14/24

Ingredients

Clarification requests are a dialogue phenomena that involves more than one utterance by more than one speaker Look at clarifications from “dialogic” and “classical linguistic” viewpoints The following topics have to be investigated:

Miscommunication (Grounding, partiality)

EGK Colloquium January 28,2002 – p.14/24

Ingredients

Clarification requests are a dialogue phenomena that involves more than one utterance by more than one speaker Look at clarifications from “dialogic” and “classical linguistic” viewpoints The following topics have to be investigated:

Miscommunication (Grounding, partiality) (Phonetic)/syntax/semantics (parallelism) of clarification questions

EGK Colloquium January 28,2002 – p.14/24

Ingredients

Clarification requests are a dialogue phenomena that involves more than one utterance by more than one speaker Look at clarifications from “dialogic” and “classical linguistic” viewpoints The following topics have to be investigated:

Miscommunication (Grounding, partiality) (Phonetic)/syntax/semantics (parallelism) of clarification questions Focus in clarification requests

EGK Colloquium January 28,2002 – p.14/24

Miscommunication (I)

EGK Colloquium January 28,2002 – p.15/24

Miscommunication (I)

Miscommunication is the source/reason for clarification

EGK Colloquium January 28,2002 – p.15/24

Miscommunication (I)

Miscommunication is the source/reason for clarification Analysis/classification of miscommunication helps to understand calcification requests

EGK Colloquium January 28,2002 – p.15/24

Miscommunication (I)

Miscommunication is the source/reason for clarification Analysis/classification of miscommunication helps to understand calcification requests Relation miscommunication ↔ clarification requests

EGK Colloquium January 28,2002 – p.15/24

Miscommunication (I)

Miscommunication is the source/reason for clarification Analysis/classification of miscommunication helps to understand calcification requests Relation miscommunication ↔ clarification requests Hirst et al. 94:

EGK Colloquium January 28,2002 – p.15/24

Miscommunication (I)

Miscommunication is the source/reason for clarification Analysis/classification of miscommunication helps to understand calcification requests Relation miscommunication ↔ clarification requests Hirst et al. 94:

misunderstanding (self/other)

EGK Colloquium January 28,2002 – p.15/24

Miscommunication (I)

Miscommunication is the source/reason for clarification Analysis/classification of miscommunication helps to understand calcification requests Relation miscommunication ↔ clarification requests Hirst et al. 94:

misunderstanding (self/other) non-understanding (none or several interpretations)

EGK Colloquium January 28,2002 – p.15/24

Miscommunication (I)

Miscommunication is the source/reason for clarification Analysis/classification of miscommunication helps to understand calcification requests Relation miscommunication ↔ clarification requests Hirst et al. 94:

misunderstanding (self/other) non-understanding (none or several interpretations) misconception (conflict with prior knowledge)

EGK Colloquium January 28,2002 – p.15/24

Miscommunication (II)

EGK Colloquium January 28,2002 – p.16/24

Miscommunication (II)

More fine-grained analysis of miscommunication that relates to grounding, types (sem. readings) of clarification

EGK Colloquium January 28,2002 – p.16/24

Miscommunication (II)

More fine-grained analysis of miscommunication that relates to grounding, types (sem. readings) of clarification Grounding (Clark ’96)

EGK Colloquium January 28,2002 – p.16/24

Miscommunication (II)

More fine-grained analysis of miscommunication that relates to grounding, types (sem. readings) of clarification Grounding (Clark ’96)

(State 0: B hasn’t noticed that A has executed some communicative behavior)

EGK Colloquium January 28,2002 – p.16/24

Miscommunication (II)

More fine-grained analysis of miscommunication that relates to grounding, types (sem. readings) of clarification Grounding (Clark ’96)

(State 0: B hasn’t noticed that A has executed some communicative behavior) State 1: B has noticed that A has executed a presentation

EGK Colloquium January 28,2002 – p.16/24

Miscommunication (II)

More fine-grained analysis of miscommunication that relates to grounding, types (sem. readings) of clarification Grounding (Clark ’96)

(State 0: B hasn’t noticed that A has executed some communicative behavior) State 1: B has noticed that A has executed a presentation State 2: B has identified A’s presentation correctly

EGK Colloquium January 28,2002 – p.16/24

Miscommunication (II)

More fine-grained analysis of miscommunication that relates to grounding, types (sem. readings) of clarification Grounding (Clark ’96)

(State 0: B hasn’t noticed that A has executed some communicative behavior) State 1: B has noticed that A has executed a presentation State 2: B has identified A’s presentation correctly State 3: B has understood what A meant by his utterance

EGK Colloquium January 28,2002 – p.16/24

Miscommunication (II)

More fine-grained analysis of miscommunication that relates to grounding, types (sem. readings) of clarification Grounding (Clark ’96)

(State 0: B hasn’t noticed that A has executed some communicative behavior) State 1: B has noticed that A has executed a presentation State 2: B has identified A’s presentation correctly State 3: B has understood what A meant by his utterance (State 4: B considers taking up A’s proposed joint project)

EGK Colloquium January 28,2002 – p.16/24

Grounding

EGK Colloquium January 28,2002 – p.17/24

Grounding

Clarification Requests indicate that certain parts of an utterance could not be grounded

EGK Colloquium January 28,2002 – p.17/24

Grounding

Clarification Requests indicate that certain parts of an utterance could not be grounded Clark: Different parts of an utterance can be on different levels of grounding

EGK Colloquium January 28,2002 – p.17/24

Grounding

Clarification Requests indicate that certain parts of an utterance could not be grounded Clark: Different parts of an utterance can be on different levels of grounding What happens to the material that is not clarified explicitly?

EGK Colloquium January 28,2002 – p.17/24

Grounding

Clarification Requests indicate that certain parts of an utterance could not be grounded Clark: Different parts of an utterance can be on different levels of grounding What happens to the material that is not clarified explicitly? Can we assume that it is already grounded?

EGK Colloquium January 28,2002 – p.17/24

Grounding

Clarification Requests indicate that certain parts of an utterance could not be grounded Clark: Different parts of an utterance can be on different levels of grounding What happens to the material that is not clarified explicitly? Can we assume that it is already grounded?

Functional relevance: Do we need a detailed model of grounding for analyzing clarification requests?

EGK Colloquium January 28,2002 – p.17/24

Grounding

Clarification Requests indicate that certain parts of an utterance could not be grounded Clark: Different parts of an utterance can be on different levels of grounding What happens to the material that is not clarified explicitly? Can we assume that it is already grounded?

Functional relevance: Do we need a detailed model of grounding for analyzing clarification requests?

(Or) Does a difference in (partial) grounding affect clarification requests?

EGK Colloquium January 28,2002 – p.17/24

Questions (I)

EGK Colloquium January 28,2002 – p.18/24

Questions (I)

Groenendijk & Stokhof ’84 possible world semantics for questions

EGK Colloquium January 28,2002 – p.18/24

Questions (I)

Groenendijk & Stokhof ’84 possible world semantics for questions (Extensional) semantic type of questions is the same as for propositions: ⟨s, t⟩

EGK Colloquium January 28,2002 – p.18/24

Questions (I)

Groenendijk & Stokhof ’84 possible world semantics for questions (Extensional) semantic type of questions is the same as for propositions: ⟨s, t⟩ Difference to propositions lies in intension: ⟨s, ⟨s, t⟩⟩ is constant for propositions but not for questions

EGK Colloquium January 28,2002 – p.18/24

Questions (I)

Groenendijk & Stokhof ’84 possible world semantics for questions (Extensional) semantic type of questions is the same as for propositions: ⟨s, t⟩ Difference to propositions lies in intension: ⟨s, ⟨s, t⟩⟩ is constant for propositions but not for questions Questions are analyzed in terms of their possible answers:

?(ϕ) → ϕ ¬ϕ ?x(ϕ(x)) →

¬∃x(ϕ(x))

. . .

∀x(ϕ(x))

EGK Colloquium January 28,2002 – p.18/24

Questions (II)

EGK Colloquium January 28,2002 – p.19/24

Questions (II)

Clausal clarifications:

A: Did Bill phone? ?(phone(b)) B: Bill? ?(ask(∧phone(b)) B: Did who phone? ?x(ask(∧phone(x)))

EGK Colloquium January 28,2002 – p.19/24

Questions (II)

Clausal clarifications:

A: Did Bill phone? ?(phone(b)) B: Bill? ?(ask(∧phone(b)) B: Did who phone? ?x(ask(∧phone(x)))

Constituent clarifications:

A: Did Bill phone? ?(phone(b)) B: Bill? ?x(x = b) B: Who is Bill ?x(x = b)

EGK Colloquium January 28,2002 – p.19/24

Questions (II)

Clausal clarifications:

A: Did Bill phone? ?(phone(b)) B: Bill? ?(ask(∧phone(b)) B: Did who phone? ?x(ask(∧phone(x)))

Constituent clarifications:

A: Did Bill phone? ?(phone(b)) B: Bill? ?x(x = b) B: Who is Bill ?x(x = b)

(Need Aloni ’99 style extension of G&S semantics to account for constituent clarifications)

EGK Colloquium January 28,2002 – p.19/24

Focus (I)

EGK Colloquium January 28,2002 – p.20/24

Focus (I)

Interpretation of focus in clarification requests is crucial to get their semantics right

EGK Colloquium January 28,2002 – p.20/24

Focus (I)

Interpretation of focus in clarification requests is crucial to get their semantics right Is there a relation between ellipsis ↔ focus ↔ reading?

A: John loves Mary B: JOHN loves Mary? (pref. confirmation?) B: JOHN? (pref. reference?)

EGK Colloquium January 28,2002 – p.20/24

Focus (I)

Interpretation of focus in clarification requests is crucial to get their semantics right Is there a relation between ellipsis ↔ focus ↔ reading?

A: John loves Mary B: JOHN loves Mary? (pref. confirmation?) B: JOHN? (pref. reference?)

What is the relation to grounding?

EGK Colloquium January 28,2002 – p.20/24

Focus (I)

Interpretation of focus in clarification requests is crucial to get their semantics right Is there a relation between ellipsis ↔ focus ↔ reading?

A: John loves Mary B: JOHN loves Mary? (pref. confirmation?) B: JOHN? (pref. reference?)

What is the relation to grounding?

Do deaccented and/or omitted parts indicate grounding?

EGK Colloquium January 28,2002 – p.20/24

Focus (II)

EGK Colloquium January 28,2002 – p.21/24

Focus (II)

Rooth ’92 theory of focus interpretation

EGK Colloquium January 28,2002 – p.21/24

Focus (II)

Rooth ’92 theory of focus interpretation Associate to each “ordinary” semantic value an additional focus semantic value

EGK Colloquium January 28,2002 – p.21/24

Focus (II)

Rooth ’92 theory of focus interpretation Associate to each “ordinary” semantic value an additional focus semantic value Should also work for questions:

Does John love ‖Mary‖F ?

‖ ‖^O =?(love(j, m)), ‖ ‖^F = {?(love(j, x))|x ∈ E}

EGK Colloquium January 28,2002 – p.21/24

Focus (II)

Rooth ’92 theory of focus interpretation Associate to each “ordinary” semantic value an additional focus semantic value Should also work for questions:

Does John love ‖Mary‖F ?

‖ ‖^O =?(love(j, m)), ‖ ‖^F = {?(love(j, x))|x ∈ E} Clausal clarifications:

A: John loves Mary B: JOHN loves Mary?

‖ ‖^O =?(ask(^∧love(j, m))), ‖ ‖^F = {?(ask(^∧love(x, m)))|x ∈ E}

EGK Colloquium January 28,2002 – p.21/24

Summary (I)

EGK Colloquium January 28,2002 – p.22/24

Summary (I)

Refinement of terms: “Repair Strategies” has changed to an analysis of clarification requests

EGK Colloquium January 28,2002 – p.22/24

Summary (I)

Refinement of terms: “Repair Strategies” has changed to an analysis of clarification requests Clarification requests are a common phenomena in dialogues (esp. in “advanced” TOD but also in the BNC (Purver et al. ’01))

EGK Colloquium January 28,2002 – p.22/24

Summary (I)

Refinement of terms: “Repair Strategies” has changed to an analysis of clarification requests Clarification requests are a common phenomena in dialogues (esp. in “advanced” TOD but also in the BNC (Purver et al. ’01)) Meta-linguistic abilities crucial for successful (natural) dialogue interaction (Perlis et al. ’98)

EGK Colloquium January 28,2002 – p.22/24

Summary (I)

Refinement of terms: “Repair Strategies” has changed to an analysis of clarification requests Clarification requests are a common phenomena in dialogues (esp. in “advanced” TOD but also in the BNC (Purver et al. ’01)) Meta-linguistic abilities crucial for successful (natural) dialogue interaction (Perlis et al. ’98) Previous work focussed mainly on syntactic analysis (ellipsis) within a HPSG framework

EGK Colloquium January 28,2002 – p.22/24

Summary (II)

EGK Colloquium January 28,2002 – p.23/24

Summary (II)

Ingredients for an analysis of clarification requests are:

EGK Colloquium January 28,2002 – p.23/24

Summary (II)

Ingredients for an analysis of clarification requests are:

a theory/classification of miscommunication (Hirst et al. ’94)

EGK Colloquium January 28,2002 – p.23/24

Summary (II)

Ingredients for an analysis of clarification requests are:

a theory/classification of miscommunication (Hirst et al. ’94) a (semantic) theory of questions (Groenendijk & Stokhof ’84)

EGK Colloquium January 28,2002 – p.23/24

Summary (II)

Ingredients for an analysis of clarification requests are:

a theory/classification of miscommunication (Hirst et al. ’94) a (semantic) theory of questions (Groenendijk & Stokhof ’84) a (semantic) theory of focus (Rooth ’92)

EGK Colloquium January 28,2002 – p.23/24

Summary (II)

Ingredients for an analysis of clarification requests are:

a theory/classification of miscommunication (Hirst et al. ’94) a (semantic) theory of questions (Groenendijk & Stokhof ’84) a (semantic) theory of focus (Rooth ’92) To keep in mind:

EGK Colloquium January 28,2002 – p.23/24

Summary (II)

Ingredients for an analysis of clarification requests are:

a theory/classification of miscommunication (Hirst et al. ’94) a (semantic) theory of questions (Groenendijk & Stokhof ’84) a (semantic) theory of focus (Rooth ’92) To keep in mind:

Special kinds of clarification: Action, location, . . .

EGK Colloquium January 28,2002 – p.23/24

Summary (II)

Ingredients for an analysis of clarification requests are:

a theory/classification of miscommunication (Hirst et al. ’94) a (semantic) theory of questions (Groenendijk & Stokhof ’84) a (semantic) theory of focus (Rooth ’92) To keep in mind:

Special kinds of clarification: Action, location, . . . Implementation and usability results in prototype dialogue system

EGK Colloquium January 28,2002 – p.23/24

What to do next?

EGK Colloquium January 28,2002 – p.24/24

What to do next?

Further investigations of “ingredients” (esp. focus, audio material)

EGK Colloquium January 28,2002 – p.24/24

What to do next?

Further investigations of “ingredients” (esp. focus, audio material) Parallelism constraints/detection of clarification: How can a system find out whether a clarification has occured?

EGK Colloquium January 28,2002 – p.24/24

What to do next?

Further investigations of “ingredients” (esp. focus, audio material) Parallelism constraints/detection of clarification: How can a system find out whether a clarification has occured? Directions to go (special topics)

EGK Colloquium January 28,2002 – p.24/24

What to do next?

Further investigations of “ingredients” (esp. focus, audio material) Parallelism constraints/detection of clarification: How can a system find out whether a clarification has occured? Directions to go (special topics)

Analysis of reference ambiguity clarifications (“Which color?”)

EGK Colloquium January 28,2002 – p.24/24

What to do next?

Further investigations of “ingredients” (esp. focus, audio material) Parallelism constraints/detection of clarification: How can a system find out whether a clarification has occured? Directions to go (special topics)

Analysis of reference ambiguity clarifications (“Which color?”)

→reasoning needed (what are sensible clarification requests?)

EGK Colloquium January 28,2002 – p.24/24

What to do next?

Further investigations of “ingredients” (esp. focus, audio material) Parallelism constraints/detection of clarification: How can a system find out whether a clarification has occured? Directions to go (special topics)

Analysis of reference ambiguity clarifications (“Which color?”)

→reasoning needed (what are sensible clarification requests?)

(Non-linguistic) action/location clarifications

EGK Colloquium January 28,2002 – p.24/24

What to do next?

Further investigations of “ingredients” (esp. focus, audio material) Parallelism constraints/detection of clarification: How can a system find out whether a clarification has occured? Directions to go (special topics)

Analysis of reference ambiguity clarifications (“Which color?”)

→reasoning needed (what are sensible clarification requests?)

(Non-linguistic) action/location clarifications

→theory of action representation/planning

EGK Colloquium January 28,2002 – p.24/24